Author's response to reviews

Title: Borderline gestational diabetes mellitus and pregnancy outcomes

Authors:

Hong Ju (hong.ju@adelaide.edu.au)
Alice R Rumbold (alice.rumbold@menzies.edu.au)
Kristyn J Willson (kristyn.willson@adelaide.edu.au)
Caroline A Crowther (caroline.crowther@adelaide.edu.au)

Version: 3 Date: 11 June 2008

Author's response to reviews: see over
Dear Editor of BMC Pregnancy and Childbirth,

RE: MS: 1925629563735532
Borderline gestational diabetes mellitus and pregnancy outcomes
Hong Ju, Alice R Rumbold, Kristyn J Willson and Caroline A Crowther

After submitting our revised manuscript on the 6\textsuperscript{th} of June 2008, we have noticed 3 minor formatting/typographical changes that need to be made.

We have made the changes listed below:

Results, Paragraph 4, Sentence 2, first “adjusted” is now “unadjusted”.

Results, Paragraph 4, Sentence 3, first “adjusted” is now “unadjusted”.

Results, Paragraph 5, Sentence 2, “adjusted” is now “unadjusted”.

Please find attached the revised manuscript including these changes.

Thank You
Caroline Crowther